Author's response to reviews

Title: Blood pressure and body mass index in an ethnically diverse sample of adolescents in Paramaribo, Suriname

Authors:

Charles Agyemang (c.o.agyemang@amc.uva.nl)
Ellen Oudeman (elineoudeman@hotmail.com)
Wilco Zijlmans (cwrzijlmans@gmail.com)
JF(Hans) Wendte (j.f.wendte@amc.uva.nl)
Karien Stronks (k.stronks@amc.uva.nl)

Version: 4 Date: 18 May 2009

Author's response to reviews: see over
17/03/09

Dear Editor

Thanks for accepting our paper for publication. Please find our revised manuscript ‘Blood pressure and body mass index in an ethnically diverse sample of adolescents in Paramaribo, Suriname’.

All the formatting changes requested have been conformed to accordingly as indicated below:

Major
1. The list of authors in the manuscript has been written exactly as they are in the submission system.
2. Reference 24 is still in press
3. Reference 32 has been cited in the manuscript text
4. The image file has excluded the title or figure number
5. Figure 2 has been submitted as one page
6. Table shading has been removed
7. Table size has been fitted into an A4 portrait page

Minor
1. The authors’ qualifications have been removed
2. The running title and key words have been removed

We hope that you would consider our manuscript for publication in its current form and we look forward to hearing from you.

Yours truly,
Dr C. Agyemang*, Miss E. Oudeman, Dr W. Zijlmans, Dr .F. Wendte, Prof. K. Stronks

* Correspondence to: Dr Charles Agyemang, Department. of Social Medicine, Academic Medical Centre, University of Amsterdam, Meibergdreef 9, 1105 AZ Amsterdam, The Netherlands, Tel: 0031 20 5664885, Fax: 0031 20 6972316
Email: c.o.agyemang@amc.uva.nl